Author’s response to reviews

Title: Using joint models to disentangle intervention effect types and baseline confounding: an application within an intervention study in prodromal Alzheimer’s disease with Fortasyn Connect

Authors:

Floor van Oudenhoven (floor.van-oudenhoven@nutricia.com; f.vanoudenhoven@erasusmc.nl)

Sophie Swinkels (Sophie.Swinkels@danone.com)

Tobias Hartmann (Tobias.Hartman@uks.eu)

Hilkka Soininen (hilkka.soininen@uef.fi)

Anneke van Hees (anneke.eijgelaar@nutricia.com)

Dimitris Rizopoulos (d.rizopoulos@erasusmc.nl)

Version: 2 Date: 16 Apr 2019

Author’s response to reviews:

Dear Dr. Sheng Luo,

Please find enclosed the revised manuscript (after a minor revision) entitled ‘Using joint models to disentangle intervention effect types and baseline confounding: an application within an intervention study in prodromal Alzheimer’s disease with Fortasyn Connect’ (BMRM-D-18-00542). We thank the reviewers again for their thoughtful critiques and valuable comments of the manuscript and we now also incorporated the last minor comment, which was to cite two-author references in a consistent manner throughout the paper.

We hope that you will find the manuscript now suitable for publication in BMC medical research methodology. If you have any questions, please do not hesitate to contact me.

Sincerely,

Floor van Oudenhoven